Author’s response to reviews

Title: Development of a tailored strategy to improve postpartum hemorrhage guideline adherence

Authors:

Suzan de Visser (suzan.devisser@radboudumc.nl)
Mallory Woiski (Mallory.Woiski@radboudumc.nl)
Richard Grol (Richard.Grol@radboudumc.nl)
Frank Vandenbussche (Frank.Vandenbussche@radboudumc.nl)
Marlies Hulscher (Marlies.Hulscher@radboudumc.nl)
Hubertina Scheepers (hcj.scheepers@mumc.nl)
Rosella Hermens (Rosella.Hermens@radboudumc.nl)

Version: 3 Date: 13 Jan 2018

Author’s response to reviews:

1) Thank you for providing an ‘Availability of data and materials’ subheading. However, we cannot currently accept the current statement provided as it does not meet our editorial policies (see http://www.biomedcentral.com/getpublished/editorial-policies#DataandMaterialRelease). Submission of a manuscript to a BioMed Central journal implies that materials described in the manuscript, including all relevant raw data, will be freely available to any scientist wishing to use them for non-commercial purposes, without breaching participant confidentiality.

Please detail where the raw data supporting your findings can be found (including information found in any supplementary files). If the raw data is publicly available or can be requested, please state that this is the case, and where the data can be found/requested from. Alternatively, if you do not wish to/cannot share your data, please state (in the ‘Availability of data and Materials’ section) that data will not be shared, and state the reason. Please note that participant or patient data should be de-identified. Please see our submission guidelines for the statements we can accept in this section (https://bmcpregnancychildbirth.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article).

Answer: We have changed our statement to the following statement: The datasets analyzed during the current study are available from the corresponding author on reasonable request. All
raw data will be publicly available through the university’s system once the university policy is finalized. The university policy and depository are currently (January 2018) still under development but will be finalized shortly.

2) Please amend your ‘Details of Ethics Approval’ subheading to ‘Ethics approval and consent to participate’ and include a statement whether informed consent, written or verbal, was obtained from all participants.

If consent was not required, please provide details of why this was waived and the institution/committee that approved the waiver (if applicable) OR state any national guidelines which waive the need for consent.

Answer: We have adjusted the heading and added information about the informed consent which was received for all involved patients in writing.

3) At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Answer: A clean version of the manuscript is made and uploaded. Figures will be attached separately.